Author's response to reviews

Title: A case report of Merkel cell carcinoma on chronic lymphocytic leukemia: differential diagnosis of coexisting lymphadenopathy and indications for early aggressive treatment

Authors:

Konstantinos I Papageorgiou MD (papageorgiou@doctors.org.uk)
Magdalini MG Kaniorou-Larai MD (mkaniorou@yahoo.co.uk)

Version: 4 Date: 17 August 2005

Author's response to reviews:

Dear Editor,

We are re-submitting the final version of the manuscript.

The following changes have been made:

1) The title includes the "case report"
2) The heading authors has been removed
3) The abstract has been re-structured in Background, Case presentation and Conclusions
4) Sentence case has been used in all headings and sub-headings
5) The manuscript includes the section Background (page 3-6)
6) The abbreviations have been specified
7) The authors have been updated

Thanks for all the kind supervision of our work

Kind Regards

Yours sincerely

The Authors